Author’s response to reviews

Title: The Immunogenicity, Safety, and Consistency of an Indonesia Combined DTP-HB-Hib Vaccine in Expanded Program on Immunization Schedule

Authors:

Kusnandi Rusmil (kusnandi@hotmail.com)
Hartono Gunardi (hartono@ikafkui.net)
Eddy Fadlyana (edfadlyana@yahoo.com)
Soedjatmiko Soedjatmiko (soedjat3@yahoo.com)
Meita Dhamayanti (meita_d@yahoo.com)
Rini Sekartini (rsekartini@yahoo.com)
Hindra Irawan Satari (hsatari54@yahoo.com)
Nelly Amalia Risan (nellyrisan@yahoo.com)
Dwi Prasetyo (d_prass2000@yahoo.com)
Rodman Tarigan (rodmantarigan@yahoo.co.id)
Reni Ghrahani (renighrahani@yahoo.com)
Mia Milanti Dewi (miamilanti@gmail.com)
Novilia Sjafri Bachtiar (novilia@biofarma.co.id)
Rini Mulia Sari (rini.mulia@biofarma.co.id)

Version: 9
Date: 7 November 2015

Author’s response to reviews: see over
Dear Catherine Olino
Journal Editorial Office
BioMed Central

Thank you for your e-mail about some suggestions of editorial revisions. We have listed all changes that we have made below:

1. Each center which approved the study protocol has been stated in Methods section. The sentences in line 83-87 has been changed to: “The study was conducted in accordance with the Declaration of Helsinki and Good Clinical Practice guidelines with approval of ethical review committees by Health Research Ethics Committee Faculty of Medicine University of Indonesia – Dr. Cipto Mangunkusumo Hospital and Health Research Ethics Committee Faculty of Medicine Padjajaran University – Dr. Hasan Sadikin Hospital.”

2. The Financial Disclosure section has been removed and the funding details have been changed and moved to Acknowledgements section. Line 416: “Bio Farma was the funding source of this study.”

3. The remaining authors did not have any competing interests to declare. This has been added in Conflict of Interests section. Line 401-402: “The remaining authors did not have any competing interests in this study.”

4. Each author’s name has been replaced with their initials and the statement "Novilia Sjafri Bachtiar and Rini Mulia Sari worked in Bio Farma that supported this study” has been removed from the Acknowledgements section. (Line 415)

5. There is also a little correction for the name of one author. The author’s name in Title Page “Nelly Raisan” has been changed to “Nelly Amalia Risan.”
6. In Title Page, the name of institution has been changed to “Child Health Department, Faculty of Medicine, Padjadjaran University / Hasan Sadikin Hospital, Bandung” and “Child Health Department, Faculty of Medicine, University of Indonesia / Cipto Mangunkusumo Hospital, Jakarta”.

7. The details of authors in Title Page has been changed to a,b,c for the label of institution and the asterisks for labelling that the authors marked with asterisk sign contributed equally to the work.

Should you need any further information, please feel free to contact me. Thank you very much for all your kind help and attention.

Yours sincerely,

Prof. Kusnandi Rusmil